Author’s response to reviews

Title: Bariatric Surgery in Individuals with HIV and Type 2 Diabetes: A Case Series

Authors:

Wei Yang (yangweifall@gmail.com)
Anjali Zalin (A.Balasanthiran@imperial.ac.uk)
Mark Nelson (Mark.nelson@imperial.ac.uk)
Gianluca Bonanomi (Gianluca.Bonanomi@chelwest.nhs.uk)
James Smellie (James.Smellie@chelwest.nhs.uk)
Kevin Shotliff (Kevin.Shotliff@chelwest.nhs.uk)
Evangelos Efthimiou (Evangelos.Efthimiou@chelwest.nhs.uk)
Veronica Greener (Veronica.Greener@chelwest.nhs.uk)

Version: 1 Date: 23 Mar 2019

Author’s response to reviews:

Dear reviewers,

Many thanks for your kind advice. We have revised point by point (highlighted yellow in the manuscript) as follows:

1. The case reports should include past medical, social, environmental, family and employment history.-----added Table 1 to explain.

2. What medications was the patient receiving prior to diagnosis? Did the patient smoke, and/or consume alcohol?-----added Table 1 to explain.
3. Give detailed physical and neurological examination on admission. What was the pulse, blood pressure and temperature, on admission?-----added Table 1 to explain.

4. Give the doses of all medications including anti HIV that were given and their doses. -----added Table 1 and part of Table 2 to explain.

5. Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc) preferably in a table. -----added Table 3 to explain.

6. Give information about follow-up for at least 6 months. -----added Table 3 and part of table 2 to explain.

7. Discussion – add a paragraph at the beginning of the Discussion that summarizes the case and describes what is unique in this case compared to what is available in the literature.-----added a paragraph at the beginning of the discussion.

Best regards

Wei Yang